Author’s response to reviews

Title: Immunoglobulin abnormalities are frequent in patients with lupus nephritis

Authors:
Maria-Jose Cuadrado (mjcuadrado@yahoo.com)
Irene Calatayud (irecalata@hotmail.com)
Maria Urquizu Padilla (mariaurquizu@gmail.com)
Sonali Wijetilleka (sonaliwijetilleka@doctors.org.uk)
Sorena Kiani-Alikhan (skiani@nhs.net)
Mohammed Karim (mkarim@sidra.org)

Version: 2 Date: 22 May 2019

Author’s response to reviews:

Dr Savino Sciascia MD PhD
BMC Rheumatology

Dear Dr Sciascia,

Many thanks for this response, and the Editorial comments.

We have responded to the Editor Comments as follows:

Editor Comments:

Minor Changes:

- The Authors should clearly acknowledge among the limitations for this study that the information requested by Reviewers 2 in points 3-7 are not available.

We have included a paragraph on limitations of the study in the discussion. Page 5 lines 232-235

- Taking into account the overall balance in length of the manuscript and the international readership of the journal, the description of the different roles between Clinical Immunologist
and Rheumatologist might be unnecessary as the scenarios are globally heterogenous and often overlapping. The concepts underlined in lines 225-233 can be summarised.

We have abbreviated this section. Page 5 lines 220-223

With many thanks

Very best wishes

Yousuf Karim